# ARTICLE

# **Annals of Internal Medicine**

# Extended-Duration Venous Thromboembolism Prophylaxis in Acutely III Medical Patients With Recently Reduced Mobility

# A Randomized Trial

Russell D. Hull, MBBS; Sebastian M. Schellong, MD; Victor F. Tapson, MD; Manuel Monreal, MD; Meyer-Michel Samama, MD, PharmD; Philippe Nicol, PhD; Eric Vicaut, MD, PhD; Alexander G.G. Turpie, MD; and Roger D. Yusen, MD, MPH, for the EXCLAIM (Extended Prophylaxis for Venous ThromboEmbolism in Acutely III Medical Patients With Prolonged Immobilization) study\*

**Background:** Extended-duration low-molecular-weight heparin has been shown to prevent venous thromboembolism (VTE) in high-risk surgical patients.

**Objective:** To evaluate the efficacy and safety of extendedduration enoxaparin thromboprophylaxis in acutely ill medical patients.

**Design:** Randomized, parallel, placebo-controlled trial. Randomization was computer-generated. Allocation was centralized. Patients, caregivers, and outcome assessors were blinded to group assignment. (ClinicalTrials.gov registration number: NCT00077753)

Setting: 370 sites in 20 countries across North and South America, Europe, and Asia.

**Patients:** Acutely ill medical patients 40 years or older with recently reduced mobility (bed rest or sedentary without [level 1] or with [level 2] bathroom privileges). Eligibility criteria for patients with level 2 immobility were amended to include only those who had additional VTE risk factors (age >75 years, history of VTE, or active or previous cancer) after interim analyses suggested lower-than-expected VTE rates.

**Intervention:** Enoxaparin, 40 mg/d subcutaneously (2975 patients), or placebo (2988 patients), for  $28 \pm 4$  days after receiving openlabel enoxaparin for an initial  $10 \pm 4$  days. Measurements: Incidence of VTE up to day 28 and of major bleeding events up to 48 hours after the last study treatment dose.

**Results:** Extended-duration enoxaparin reduced VTE incidence compared with placebo (2.5% vs. 4%; absolute risk difference favoring enoxaparin, -1.53% [95.8% CI, -2.54% to -0.52%]). Enoxaparin increased major bleeding events (0.8% vs. 0.3%; absolute risk difference favoring placebo, 0.51% [95% CI, 0.12% to 0.89%]). The benefits of extended-duration enoxaparin seemed to be restricted to women, patients older than 75 years, and those with level 1 immobility.

**Limitation:** Estimates of efficacy and safety for the overall trial population are difficult to interpret because of the change in eligibility criteria during the trial.

**Conclusion:** Use of extended-duration enoxaparin reduces VTE more than it increases major bleeding events in acutely ill medical patients with level 1 immobility, those older than 75 years, and women.

Primary Funding Source: Sanofi-aventis.

www.annals.org).

Ann Intern Med. 2010;153:8-18. www.annals.org For author affiliations, see end of text. \* For a list of the EXCLAIM study investigators, see Appendix 1 (available at

ospitalized medical patients have a high risk for venous thromboembolism (VTE) (1). Recent studies (2-4) suggest that 5% to 15% of medical patients who do not receive appropriate prophylaxis develop objectively confirmed deep venous thrombosis (DVT). Up to 75% of fatal pulmonary embolism cases occur in hospitalized medical patients (5), and VTE is associated with considerable

#### See also:

# Print

Editors' Notes	. 9
Editorial comment	51
Summary for Patients	50

# Web-Only

Appendixes	
Appendix Tables	
Appendix Figures	
Conversion of graphics into slides	

8 © 2010 American College of Physicians

long-term morbidity and substantial consumption of hospital resources (6).

Large clinical trials have demonstrated the benefits of short-term VTE prophylaxis in hospitalized acutely ill medical patients; these benefits are mostly attributable to reductions in asymptomatic VTE (2-4). On the basis of this evidence, guidelines recommend use of unfractionated heparin, low-molecular-weight heparin, or fondaparinux for medical patients with heart failure or severe respiratory disease or those who are confined to bed with 1 or more risk factors, such as cancer, previous VTE, or inflammatory bowel disease (1).

Extended-duration (4-week) prophylaxis has been shown to significantly reduce the incidence of VTE compared with a standard (1-week) regimen in high-risk surgical patients (7–10). However, studies have not assessed the efficacy and safety of extended-duration prophylaxis in acutely ill medical patients. We hypothesized that an extended-duration enoxaparin regimen similar to that evaluated in patients undergoing elective hip arthroplasty (7) would be beneficial for acutely ill medical patients at high risk for VTE (11). We designed a randomized clinical trial to assess the efficacy and safety of extended-duration enoxaparin compared with placebo, following a 6- to 14-day, clinically proven regimen for VTE prophylaxis (2, 11).

# **METHODS**

This was an international, multicenter, parallelgroup, double-blind trial (11). We conducted it according to standards of good clinical practice and in keeping with the Declaration of Helsinki and local regulations. The institutional review board at each site approved the trial. Participation required written informed patient consent.

# Setting and Patients

We enrolled patients from 370 hospitals across 20 countries between February 2002 and March 2006 (Appendix 1, available at www.annals.org). We recruited patients with acute medical illness (for example, heart failure, respiratory insufficiency, or infection) who met previously reported eligibility criteria (11). In brief, patients were eligible if they were at least 40 years of age, had a life expectancy of at least 6 months, and had recently reduced mobility for up to 3 days. In addition, they had to be considered by the enrolling investigator as likely to have reduced mobility for at least 3 days after enrollment. We defined "reduced mobility" as requiring total bed rest or being sedentary without bathroom privileges (level 1 immobility) or with bathroom privileges (level 2 immobility). We subsequently amended the eligibility criteria as described in the Trial Monitoring and Amendment section.

# Randomization and Interventions

Enrolled patients received open-label subcutaneous enoxaparin, 40 mg/d, for  $10 \pm 4$  days; some completed open-label prophylaxis in the outpatient setting. Patients who successfully completed open-label prophylaxis were randomly assigned in a 1:1 ratio in a double-blind manner to receive either subcutaneous enoxaparin, 40 mg/d, or placebo for an additional  $28 \pm 4$  days. The random assignment list was computer-generated in permuted blocks of 4, stratified by center, by a clinical research organization (Covance, Radnor, Pennsylvania); staff obtained allocation assignments at the time of treatment by using a telephone interactive voice-response system.

During hospitalization, authorized study personnel administered the study drug. In the outpatient setting, the study drug was administered by study staff, health care providers, or patients (who had successfully completed training in the proper administration technique).

# Efficacy Outcomes

Our primary efficacy end point was VTE, defined as the composite of symptomatic or asymptomatic proximal DVT, symptomatic pulmonary embolism, or fatal pulmonary embolism, during the double-blind treatment period ( $28 \pm 4$  days after random assignment). Secondary end points were VTE incidence through 3 months (day 90  $\pm$  10) and mortality at 1 month (day 28  $\pm$  4), 3

#### Context

Four weeks of enoxaparin therapy reduces VTE incidence more than 1 week of treatment in surgical patients at high risk for VTE. The same has not yet been shown for medical patients.

# Contribution

Adding 28 days of enoxaparin treatment to an initial 10day course reduced VTE incidence more than it increased major bleeding events in female, older, or sedentary patients with acute medical illness.

# Caution

Trial eligibility criteria had to be modified after interim analyses suggested that extended-duration enoxaparin did more harm than good.

# Implication

Extended-duration enoxaparin seems to have a favorable benefit-risk ratio in high-risk subgroups of patients with acute medical illness.

—The Editors

months (day 90  $\pm$  10), and 6 months (day 180  $\pm$  10). We assessed all efficacy end points, except mortality, in randomly assigned patients who received at least 1 dose of the study treatment during the double-blind treatment period and had at least 1 interpretable ultrasonogram during the period or up to 7 days after (efficacy population).

We used bilateral compression ultrasonography or venography to evaluate patients with suspected DVT on the basis of symptoms during the double-blind treatment period. We used computed tomography or ventilationperfusion lung scanning to evaluate suspected symptomatic cases of pulmonary embolism. When available, we used autopsy results to assess the presence of fatal pulmonary emboli. We counted each symptomatic event as a primary efficacy end point only if sufficient data were available for confirmation by central adjudication.

At the end of the double-blind treatment period, patients underwent bilateral ultrasonography of the lower extremities to identify asymptomatic proximal DVT. Locally trained and certified operators used a standardized B-mode compression ultrasonography protocol that required 10 scans at predefined segments of the femoral and popliteal veins. In addition, all scans were centrally read by reviewers at Bio-Imaging Technologies (Newtown, Pennsylvania), who were blinded to treatment assignment, clinical information, and the local interpretation of the scan. Discrepancies between local and central readings were centrally adjudicated in blinded sessions (**Appendix 2**, available at www.annals.org) (11).

# Safety Outcomes

The primary safety end point was the incidence of major hemorrhagic complications, during and up to 48

**ARTICLE** | Extended-Duration Venous Thromboembolism Prophylaxis in Medical Patients

Figure 1. Study flow diagram. Enrolled (n = 7500 [5536; 1964]) Excluded (n = 85 [63; 22]) Adverse events: 2 (2; 0) Lost to follow-up: 5 (5; 0) Died: 4 (2:2) Withdrew consent: 38 (26; 12) Other: 26 (24: 2) Progressive disease: 1 (0; 1) Missing data: 9 (4; 5) Received open-label prophylaxis (n = 7415 [5473; 1942]) Excluded (n = 1331 [1042; 289]) Adverse events: 312 (238: 74) Lost to follow-up: 35 (29; 6) Randomly assigned (n = 6085 [4432; Died: 70 (47; 23) 16531)\* Withdrew consent: 332 (273; Level 1: 2611 (1197: 1414) 59) Other: 512 (399: 113) High-risk level 2: 1503 (1299: 204) Low-risk level 2: 1948 (1914; 34) Progressive disease: 70 (56; 14) Enoxaparin group (n = 3034Placebo group (n = 3051 [2226; [2206; 828])\* 8251)\* Level 1: 1311 (599; 712) Level 1: 1300 (598; 702) High-risk level 2: 741 (645; 96) High-risk level 2: 762 (645; 108) Low-risk level 2: 970 (951: 19) Low-risk level 2: 978 (963: 15) Excluded (n = 59 [47; 12]) Excluded (n = 63 [50; 13]) Adverse events: 3 (2; 1) Adverse events: 2 (1; 1) Lost to follow-up: 5 (3; 2) Lost to follow-up: 14 (10; 4) Died: 7 (4: 3) Died: 4 (3: 1) Withdrew consent: 17 (11; 6) Withdrew consent: 15 (13; 2) Other: 10 (10; 0) Other: 10 (9; 1) Progressive disease: 0 (0: 0) Progressive disease: 1 (1; 0) Missing data: 20 (18; 2) Missing data: 14 (12; 2) Safety population Received placebo (n = 2988 [2176; Received enoxaparin (n = 2975 [2159: 816]) 8121) Level 1: 1281 (589; 692) Level 1: 1292 (590; 702) High-risk level 2: 721 (627; 94) High-risk level 2: 746 (640; 106) Low-risk level 2: 951 (932; 19) Low-risk level 2: 950 (936; 14) Excluded (n = 478 [309; 169]) Excluded (n = 490 [341: 149]) Problems with ultrasonography Problems with ultrasonography 232 (139; 93)‡§ 215 (114: 101)# Withdrew: 258 (202; 56)¶ Withdrew: 263 (195; 68)\*\* population ++ Received enoxaparin (n = 2485Received placebo (n = 2510[1818; 667]) [1867; 643]) Level 1: 1070 (496; 574) Level 1: 1040 (488; 552) High-risk level 2: 596 (519; 77) High-risk level 2: 641 (561; 80) Efficacy | Low-risk level 2: 809 (794; 15) Low-risk level 2: 818 (807; 11)

Numbers are reported for the total study population, followed by the number of patients enrolled before and after the eligibility criteria were amended. \* Includes 1 patient who did not receive enoxaparin during the open-label phase of the trial but received it during the double-blind phase and 23 patients (12 enoxaparin recipients and 11 placebo recipients; 22 preamendment [11 enoxaparin recipients and 11 placebo recipients], 1 postamendment [1 enoxaparin recipient]) for whom immobility classification data were missing or immobility-level classification was absent.

+ Includes 22 patients (21 preamendment [10 enoxaparin recipients and 11 placebo recipients] and 1 postamendment [enoxaparin recipient]) who had missing immobility-level data or no immobility-level classification.

**‡** Reasons include incompatible digital format, incomplete scans, and insufficient quality of scans.

§ No ultrasonography, reason not known, 65 (48; 17); evaluated outside of time window, 5 (4; 1); not sent for central reading, 43 (23; 20); not evaluable centrally, 100 (56; 44); not read centrally because of technical problems, 19 (8; 11).

No ultrasonography, reason not known, 53 (38; 15); evaluated outside of time window, 2 (2; 0); not sent for central reading, 40 (20; 20); not evaluable centrally, 96 (44; 52); not read centrally because of technical problems, 24 (10; 14).

¶ Adverse event, 76 (60; 16); lost during follow-up, 6 (5; 1); died, 25 (17; 8); discontinued study participation, 61 (52; 9); progressive disease, 14 (11; 3); other, 76 (57; 19).

\*\* Adverse event, 85 (70; 15); lost during follow-up, 13 (9; 4); died, 32 (20; 12); discontinued study participation, 56 (36; 20); progressive disease, 15 (15; 0); other, 62 (45; 17).

++ 21 patients were excluded from the immobility subgroups (20 preamendment [9 enoxaparin recipients and 11 placebo recipients] and 1 postamendment [enoxaparin recipient]) because immobility-level data were missing or immobility-level classification was absent.

Table 1. Base	eline Patient	Characteristics	at Baseline*
---------------	---------------	-----------------	--------------

Variable	Preamendment		Postamendment		Total Population	
	Extended- Duration Enoxaparin (n = 2159)	Placebo ( <i>n</i> = 2176)	Extended- Duration Enoxaparin (n = 816)	Placebo ( <i>n</i> = 812)	Extended- Duration Enoxaparin (n = 2975)	Placebo (n = 2988)
Demographic characteristics						
Mean age (SD), y	67.8 (12.2)	67.2 (12.4)	68.1 (12.0)	68.2 (12.7)	67.9 (12.1)	67.5 (12.5)
Men, n (%)	1081 (50.1)	1076 (49.4)	386 (47.3)	401 (49.4)	1467 (49.3)	1477 (49.4)
Race or ethnicity, n (%)						
White	1678 (77.7)	1685 (77.4)	548 (67.2)	551 (67.9)	2226 (74.8)	2236 (74.8)
Black	156 (7.2)	155 (7.1)	43 (5.3)	49 (6.0)	199 (6.7)	204 (6.8)
Hispanic	282 (13.1)	301 (13.8)	109 (13.4)	106 (13.1)	391 (13.1)	407 (13.6)
Asian or Oriental	17 (0.8)	15 (0.7)	69 (8.5)	66 (8.1)	86 (2.9)	81 (2.7)
Multiracial	9 (0.4)	13 (0.6)	39 (4.8)	33 (4.1)	48 (1.6)	46 (1.5)
Other	17 (0.8)	7 (0.3)	8 (1.0)	7 (0.9)	25 (0.8)	14 (0.5)
Europo	919 (12 5)	912 (11 9)	212 (20 0)	247 (20 4)	1161 (20.0)	1150 (20 0)
North America	806 (37.3)	912 (41.9) 813 (37 <i>A</i> )	164 (20.1)	165 (20.3)	970 (32.6)	978 (32.7)
Latin America	281 (13.0)	290 (13 3)	88 (10.8)	87 (10 7)	369 (12.4)	377 (12.6)
Africa Asia or the Middle Fast	17 (0.8)	19 (0.9)	309 (37 9)	303 (37 3)	326 (11.0)	322 (10.8)
Australia	137 (6.3)	142 (6.5)	12 (1.5)	10 (1.2)	149 (5.0)	152 (5.1)
Mean body mass index (SD), $kg/m^2$	28.9 (8.6)	29.0 (8.6)	28.3 (7.2)	27.9 (7.3)	28.7 (8.3)	28.7 (8.2)
Body mass index $\geq$ 30 kg/m <sup>2</sup> , n (%)	743 (34.4)	764 (35.1)	275 (33.7)	244 (30.0)	1018 (34.2)	1008 (33.7)
Primary enrollment diagnoses, n (%)						
Acute infection without septic shock	760 (35.2)	794 (36.5)	217 (26.6)	211 (26.0)	977 (32.8)	1005 (33.6)
Acute respiratory insufficiency	692 (32.1)	666 (30.6)	213 (26.1)	234 (28.8)	905 (30.4)	900 (30.1)
Heart failure‡	329 (15.2)	350 (16.1)	217 (26.6)	214 (26.4)	546 (18.4)	564 (18.9)
Post-acute ischemic stroke	131 (6.1)	127 (5.8)	67 (8.2)	64 (7.9)	198 (6.7)	191 (6.4)
Acute rheumatic disorders	59 (2.7)	62 (2.8)	16 (2.0)	21 (2.6)	75 (2.5)	83 (2.8)
Active cancer	33 (1.5)	36 (1.7)	17 (2.1)	10 (1.2)	50 (1.7)	46 (1.5)
Fracture	10 (0.5)	12 (0.6)	10 (1.2)	5 (0.6)	20 (0.7)	17 (0.6)
Multiple diagnoses	10 (0.5)	4 (0.2)	11 (1.3)	8 (1.0)	21 (0.7)	12 (0.4)
Active inflammatory bowel disease	4 (0.2)	8 (0.4)	3 (0.4)	0 (0.0)	7 (0.2)	8 (0.3)
Other	131 (6.1)	117 (5.4)	45 (5.5)	45 (5.5)	176 (5.9)	162 (5.4)
Disk factors and immobility lovel $n(\%)$						
Lovel 1 immobility	590 (27 2)	599 (27 1)	702 (96 0)	692 (95 2)	1202 (42 4)	1201 (12 0)
Level 2 immobilityS	1559 (27.3)	1576 (72 4)	113 (13.8)	120 (14.8)	1672 (45.4)	1696 (56.8)
$\Delta ge > 75 v^{**}$	431 (27.6)	423 (26.8)	69 (61 1)	78 (65.0)	500 (29 9)	501 (29.5)
Cancer**	210 (13 5)	246 (15.6)	25 (22 1)	26 (21 7)	235 (14 1)	272 (16.0)
History of VTE**	109 (7.0)	116 (7.4)	16 (14.2)	17 (14.2)	125 (7.5)	133 (7.8)
Age $>75$ v	632 (29.3)	635 (29.2)	246 (30.1)	268 (33.0)	878 (29.5)	903 (30.2)
Active or previous cancer	296 (13.7)	329 (15.1)	99 (12.1)	93 (11.5)	395 (13.3)	422 (14.1)
History of VTE	143 (6.6)	154 (7.1)	57 (7.0)	48 (5.9)	200 (6.7)	202 (6.8)
Obesity (body mass index $\geq$ 30 kg/m <sup>2</sup> )	743 (34.4)	764 (35.1)	275 (33.7)	244 (30.0)	1018 (34.2)	1008 (33.7)
Venous insufficiency	293 (13.6)	305 (14.0)	113 (13.8)	104 (12.8)	406 (13.6)	409 (13.7)
Hormone therapy	58 (2.7)	54 (2.5)	9 (1.1)	9 (1.1)	67 (2.3)	63 (2.1)
Chronic heart failure	488 (22.6)	493 (22.7)	266 (32.6)	277 (34.1)	754 (25.3)	770 (25.8)
Chronic respiratory failure	879 (40.7)	882 (40.5)	304 (37.3)	309 (38.1)	1183 (39.8)	1191 (39.9)
Chronic inflammatory disease	7 (0.3)	19 (0.9)	1 (0.1)	2 (0.2)	8 (0.3)	21 (0.7)
Family history of VTE	4 (0.2)	1 (0.0)	0	0	4 (0.1)	1 (0.0)
Thrombophilia	4 (0.2)	3 (0.1)	0	0	4 (0.1)	3 (0.1)
Draviava madiationa (0/)						
Antiplatelet or anti informations diverse	500 (27 2)	EAT (2E 4)	167 (20 5)	167 (20 0)	756 (25 4)	714 (22.0)
Antiplatelet or anti-Inflammatory drugs	203 (27.3) 151 (7.0)	547 (25.1) 155 (7.4)	167 (20.5)	107 (20.6) 50 (6.2)	756 (25.4) 206 (6.0)	7 14 (23.9)
Anti-inflammatory drugs	555 (25.7)	502 (23 1)	155 (19 0)	151 (18 6)	710 (0.9)	200 (0.9)
And-initiation utugs	555 (25.7)	502 (25.1)	155 (19.0)	(0.0)	/10(25.9)	005 (21.9)

VTE = venous thromboembolism.
\* Data are for the trial safety population of 5963 patients.
\* Europe: France, Germany, Great Britain, Italy, Poland, Russia, and Spain. North America: Canada and the United States. Latin America: Argentina, Brazil, Colombia, and Mexico. Africa, Asia, or the Middle East: India, Israel, South Africa, and Tunisia.
‡ Modified New York Heart Association functional class III or IV.
§ We excluded 22 patients in the total safety population (21 preamendment [10 from the enoxaparin group and 11 from the placebo group] and 1 postamendment [from the enoxaparin group]) because of missing immobility-level data or absence of immobility-level classification.

Total bed rest or sedentary without bathroom privileges.
 Total bed rest or sedentary with bathroom privileges.
 \*\* Patients in these subgroups could have >1 risk factor for VTE.

# Table 2. Incidence of Primary Efficacy and Safety Outcomes

End Point		Preamendment	*		Postamendmen	t
	Extended-Duration Enoxaparin, <i>n/N</i> (%)	Placebo, n/N (%)	Absolute Risk Difference (CI), %†	Extended-Duration Enoxaparin, <i>n/N (%)</i>	Placebo, n/N (%)	Absolute Risk Difference (CI), %†
VTE‡§						
All	45/1818 (2.5)	78/1867 (4.2)	-1.70 (-2.86 to -0.55)	16/667 (2.4)	22/643 (3.4)	-1.02 (-2.85 to 0.80)
Level 1	12/496 (2.4)	30/488 (6.1)	-3.73 (-6.25 to -1.20)	13/574 (2.3)	17/552 (3.1)	-0.81 (-2.70 to 1.07)
Level 2	33/1313 (2.5)	47/1368 (3.4)	-0.92 (-2.21 to 0.36)	3/92 (3.3)	5/91 (5.5)	-2.23 (-8.16 to 3.69)
High-risk∥	18/519 (3.5)	31/561 (5.5)	-2.05 (-4.51 to 0.41)	3/77 (3.9)	5/80 (6.3)	-2.35 (-9.20 to 4.49)
Low-risk¶	15/794 (1.9)	16/807 (2.0)	-0.09 (-1.44 to 1.26)	0/15 (0.0)	0/11 (0.0)	-
Maior blooding overto**						
Major bleeding events"	40 (2450 (0.0)	40/2476 (0.5)	0.42 ( 0.07 + 0.04)	C(0AC(0, 7))	0/042 (0.0)	074 (045 + 4 22)
All	19/2159 (0.9)	10/21/6 (0.5)	0.42 (-0.07 to 0.91)	6/816 (0.7)	0/812 (0.0)	0.74 (0.15 to 1.32)
Level 1	5/590 (0.8)	2/589 (0.3)	0.51 (-0.37 to 1.38)	4/702 (0.6)	0/692 (0.0)	0.57 (0.01 to 1.13)
Level 2	14/1559 (0.9)	8/1576 (0.5)	0.39 (-0.19 to 0.98)	2/113 (1.8)	0/120 (0.0)	1.77 (-0.66 to 4.20)
High-risk∥	4/627 (0.6)	3/640 (0.5)	0.17 (-0.65 to 0.99)	2/94 (2.1)	0/106 (0.0)	2.13 (-0.79 to 5.04)
Low-risk¶	10/932 (1.1)	5/936 (0.5)	0.54 (-0.27 to 1.35)	0/19 (0.0)	0/14 (0.0)	0.00 (0.00 to 0.00)

VTE = venous thromboembolism.

\* Data include outcomes that were not fully adjudicated at the time of interim analysis. The analysis using adjudicated data only found no significant difference between groups. See "Trial Monitoring and Amendment" in the Methods section for further details.

 $\stackrel{+}{+}$  For VTE end points in the total population, we report 95.8% CIs (P < 0.042) because of the  $\alpha$  adjustment for the interim analysis. For all other end points, we report 95% CIs (P < 0.050).

\* Assessed in the efficacy population (4995 patients). We excluded 21 patients from the immobility subgroups (20 preamendment [9 in the enoxaparin group and 11 in the § One patient included in the placebo group of the preamendment and total populations experienced a VTE but had missing immobility-level data or absence of immobility-level data or absence of simmobility-level data or absence of s

immobility-level classification.

Il Patients with level 2 immobility and ≥1 of the following VTE risk factors: age >75 y, history of VTE, or active or previous cancer. Il Patients with level 2 immobility and none of the additional specified risk factors for VTE. \*\* Post hoc analysis performed by using a threshold hemoglobin decrease of ≥20 g/L for a major bleeding event. Assessed in the safety population (5963 patients), except for 22 patients (21 preamendment [10 from the enoxaparin group and 11 from the placebo group] and 1 postamendment [from the enoxaparin group]) whom we excluded because of missing immobility-level data or absence of immobility-level classification. Appendix Table 3 (available at www.annals.org) describes the types of major bleeding events we observed.

hours after the double-blind treatment period, as determined by central adjudication with blinding to treatment assignment (Appendix 2). Secondary safety end points were the incidence of major and minor hemorrhagic complications, serious adverse events, and thrombocytopenia. We assessed safety end points and mortality in all randomly assigned patients who received at least 1 dose of study medication (safety population).

Hemorrhages were considered to be major if they were overt and associated with death; a decrease in hemoglobin level of at least 30 g/L or a transfusion of at least 2 units of packed red blood cells or whole blood; surgical intervention; or retroperitoneal, intracranial, or intraocular bleeding. In a post hoc analysis, we used a more stringent threshold hemoglobin decrease of 20 g/L, as used in other trials, to further assess major bleeding events.

Minor hemorrhages were those that were overt and did not meet the criteria for a major hemorrhage. These included epistaxis lasting more than 5 minutes or requiring intervention, ecchymosis or hematoma larger than 5 cm, hematuria not associated with urinary catheter trauma, subconjunctival or gastrointestinal hemorrhage, or wound hematoma. We obtained platelet counts at the end of both the open-label and double-blind treatment phases.

We defined adverse events as new illness, worsening of preexisting illness, study medication effects (including comparator), or a combination of these. Serious adverse events were those that resulted in death or persistent or

12 6 July 2010 Annals of Internal Medicine Volume 153 • Number 1

substantial disability or incapability, were life-threatening or considered an important medical event, or required inpatient hospitalization or prolongation of existing hospitalization. Bleeding events and VTE were considered serious adverse events if they met the above criteria.

# Trial Monitoring and Amendment

An independent data safety monitoring board (DSMB) (Appendix 1) conducted interim analyses of adjudicated efficacy and safety outcomes when 25%, 50%, and 75% of the target enrollment (4044 patients) had been recruited. For the final interim analysis, adjudicated efficacy data were available for 3056 of 3685 patients with evaluable ultrasonograms and adjudicated safety data were available for 4060 patients.

The final interim efficacy analysis found lower-thanassumed VTE rates, with no statistically significant difference between treatment groups (37 of 1526 [2.4%] in the enoxaparin group vs. 50 of 1530 (3.3%) in the placebo group; P = 0.16) after unblinding and an 8.67% chance of finding such a significant difference. The interim safety analysis also found a statistically significant increase in major hemorrhages associated with 1 treatment group (13 of 2020 [0.64%] in the enoxaparin group vs. 6 of 2040 [0.29%] in the placebo group; P = 0.05) after unblinding. On the basis of these analyses, the DSMB recommended that the study as designed be terminated. The DSMB also highlighted that event rates in patients with level 1 immo-

<i>Table 2</i> —Continued							
Total Population							
Extended-Duration Enoxaparin, <i>n/N (%)</i>	Placebo, n/N (%)	Absolute Risk Difference (CI), %†					
61/2485 (2.5)	100/2510 (4.0)	-1.53 (-2.54 to -0.52)					
25/1070 (2.3)	47/1040 (4.5)	-2.18 (-3.80 to -0.57)					
36/1405 (2.6)	52/1459 (3.6)	-1.00 (-2.31 to 0.31)					
21/596 (3.5)	36/641 (5.6)	-2.09 (-4.50 to 0.31)					
15/809 (1.9)	16/818 (2.0)	-0.10 (-1.48 to 1.28)					
25/2975 (0.8)	10/2988 (0.3)	0.51 (0.12 to 0.89)					
9/1292 (0.7)	2/1281 (0.2)	0.54 (0.04 to 1.04)					
16/1672 (1.0)	8/1696 (0.5)	0.49 (-0.08 to 1.05)					
6/721 (0.8)	3/746 (0.4)	0.43 (-0.37 to 1.23)					
10/951 (1.1)	5/950 (0.5)	0.53 (-0.27 to 1.32)					

bility were consistent with the study design assumptions, which suggested that the trial could be continued in this subgroup of patients. The steering committee requested further analyses to identify whether any patients with level 2 immobility had increased risk for VTE because such patients may also have had event rates consistent with initial hypotheses. The steering committee requested a blinded multivariate analysis on risk factors for VTE (performed by the sponsor because of limited DSMB resources). This analysis identified patients with level 2 immobility and age older than 75 years, previous VTE, or active or previous cancer as having increased VTE risk and event rates consistent with study design assumptions. The steering committee therefore issued a protocol amendment that changed the eligibility criteria to require patients with level 2 immobility to also have 1 or more of these risk factors (termed "high-risk level 2"). The DSMB reviewed the amended protocol and considered it a safe and feasible way to continue the study. Enrollment criteria for patients with level 1 immobility remained unchanged, and we continued to enroll patients with level 1 immobility during and after the protocol amendment process.

# Statistical Analysis

Our original statistical assumptions are described elsewhere (11). In brief, we based our sample size calculation on an assumed VTE rate of 4.0% at day 28 in the placebo group after random assignment. With 80% power and a 2-sided type I error rate of 5%, we needed 4044 evaluable patients to show a 40% reduction in the VTE incidence with the extended-duration enoxaparin regimen; with an expected evaluability rate of 70%, we aimed to enroll 5800 patients. **Appendix 2** describes event rate assumptions and modified target enrollment after the protocol amendment.

All end points were assessed in 3 study populations: preamendment (patients enrolled before the level 2 eligibility criteria amendment), postamendment (patients enrolled after the eligibility criteria amendment), and total. We performed primary efficacy and safety analyses only on adjudicated events and assessed them in secondary analyses by immobility subgroup (level 1, level 2, high-risk level 2 [ $\geq$ 1 of 3 VTE risk factors], or low-risk level 2 [0 of 3 VTE risk factors]). We also assessed selected efficacy and safety end points in patients stratified by use of antiplatelet therapy for 1 or more days and performed interaction analyses to assess the impact of antiplatelet therapy on the observed treatment effect.

We compared incidence of VTE and bleeding between randomly assigned treatment groups by using chi-square and Fisher exact tests. We performed a time-to-event analysis for all-cause mortality by using a Cox proportional hazards model, with the treatment group as the only covariate; according to our visual inspection of the curves, the proportional hazards assumption was met. After correcting for the interim analysis, we set the critical P value at 0.042 for the final analysis of the primary efficacy end point in the total population and reported 95.8% CI values. For all other end points, a P value less than 0.05 was considered significant. We conducted formal tests of interaction between the study drug and age, sex, and immobility level and constructed forest plots for the primary efficacy end point. We did not conduct formal tests of interaction for the primary safety end point because there were too few events. We performed a sensitivity analysis for the primary efficacy end point, which included all patients who withdrew or were excluded from the study after random assignment. We used SAS, version 9.1 (SAS Institute, Cary, North Carolina), for all statistical analyses.

# Role of the Funding Source

This study was sponsored by sanofi-aventis (Paris, France), which also funded editorial support for the preparation of this article. The steering committee (Appendix 1) proposed the study design and was responsible for overseeing the conduct of the study and for reviewing and interpreting the data. The study protocol was developed and written by a clinical team at sanofi-aventis on the basis of the study design proposal. Staff at sanofi-aventis performed data management and statistical analysis. The steering committee developed the manuscript, with input from the sanofi-aventis clinical team and final authority from all coauthors.

# RESULTS

#### **Baseline Characteristics and Study Groups**

Of 7500 initially enrolled patients (Figure 1), 7415 received open-label prophylaxis with enoxaparin (median duration, 8.0 days [interquartile range {IQR}, 6.0 to 10.0 days]) (Figure 1). Of 6085 patients who completed open-label therapy and were randomly assigned, 5963 received at least 1 dose of the study drug (median treatment duration, 27.0 days [IQR, 24.0 to 29.0 days] for the enoxaparin group vs. 28.0 days [IQR, 24.0 to 29.0 days] for the placebo group); these patients comprised our safety popula-

#### Table 3. Incidence of the Individual Components of the Composite VTE End Point at Days 28 and 90

End Point		Preamendmer	ıt*		Postamendme	nt
	Extended-Duration Enoxaparin (n = 1818)	Placebo ( <i>n</i> = 1867)	Absolute Risk Difference (CI), %†	Extended-Duration Enoxaparin (n = 667)	Placebo ( <i>n</i> = 643)	Absolute Risk Difference (CI), %†
Day 28, n (%)‡						
Symptomatic VTE	4 (0.2)	22 (1.2)	-0.96 (-1.49 to -0.42)	1 (0.1)	2 (0.3)	-0.16 (-0.68 to 0.36)
Proximal DVT	44 (2.4)	73 (3.9)	-1.49 (-2.62 to -0.36)	16 (2.4)	22 (3.4)	-1.02 (-2.85 to 0.80)
Symptomatic	4 (0.2)	18 (1.0)	-0.74 (-1.24 to -0.25)	1 (0.1)	2 (0.3)	-0.16 (-0.68 to 0.36)
Asymptomatic	40 (2.2)	55 (2.9)	-0.75 (-1.77 to 0.28)	15 (2.2)	20 (3.1)	-0.86 (-2.61 to 0.89)
PE§	1 (0.1)	5 (0.3)	-0.21 (-0.47 to 0.05)	0	0	-
Symptomatic	0	4 (0.2)	-0.21 (-0.43 to -0.00)	0	0	-
Asymptomatic	1 (0.1)	1 (0.1)	0.00 (-0.15 to 0.15)	0	0	-
Fatal	0	1 (0.1)	-0.05 (-0.16 to 0.05)	0	0	-
Day 90, n (%)‡						
VTE	48 (2.6)	83 (4.4)	-1.81 (-3.00 to -0.61)	17 (2.5)	22 (3.4)	-0.87 (-2.72 to 0.97)
Symptomatic VTE	7 (0.4)	26 (1.4)	-1.01 (-1.62 to -0.40)	1 (0.1)	2 (0.3)	-0.16 (-0.68 to 0.36)
Proximal DVT	45 (2.5)	76 (4.1)	-1.60 (-2.74 to -0.45)	16 (2.4)	22 (3.4)	-1.02 (-2.85 to 0.80)
Symptomatic	5 (0.3)	21 (1.1)	-0.85 (-1.39 to -0.31)	1 (0.1)	2 (0.3)	-0.16 (-0.68 to 0.36)
Asymptomatic	40 (2.2)	55 (2.9)	-0.75 (-1.77 to 0.28)	15 (2.2)	20 (3.1)	-0.86 (-2.61 to 0.89)
PE§	3 (0.2)	7 (0.4)	-0.21 (-0.54 to 0.12)	1 (0.1)	0	0.15 (-0.14 to 0.44)
Symptomatic	2 (0.1)	5 (0.3)	-0.16 (-0.44 to 0.13)	0	0	-
Asymptomatic	1 (0.1)	2 (0.1)	-0.05 (-0.24 to 0.13)	1 (0.1)	0	0.15 (-0.14 to 0.44)
Fatal	0 (0.0)	2 (0.1)	-0.11 (-0.26 to 0.04)	1 (0.1)	0	0.15 (-0.14 to 0.44)

DVT = deep venous thrombosis; PE = pulmonary embolism; VTE = venous thromboembolism.

\* Data include outcomes that were not fully adjudicated at the time of interim analysis. The analysis using adjudicated data only found no significant difference between groups. See "Trial Monitoring and Amendment" in the Methods section for further details.

<sup>+</sup> For VTE end points in the total population, we report 95.8% CIs (P < 0.042) because of the  $\alpha$  adjustment for the interim analysis. For all other end points, we report 95% CIs (P < 0.050).

# Assessed in the efficacy population (4995 patients). We excluded 21 patients from the immobility subgroups (20 preamendment [9 in the enoxaparin group and 11 in the placebo group] and 1 postamendment [from the enoxaparin group]) because of missing immobility-level data or absence of immobility-level classification.

§ Includes asymptomatic PE events in 1 patient in the placebo group (who showed no clinical symptoms of VTE before death; PE was proven on autopsy) and 1 asymptomatic event in the enoxaparin group (detected after routine examination of a patient by using spiral computed tomography).

tion. Median treatment durations were similar in patients enrolled before and after the amendment (27.0 days [IQR, 24.0 to 29.0 days] vs. 28.0 days [IQR, 25.0 to 29.0 days] for both groups). Nine hundred sixty-eight patients withdrew from the study or had unevaluable ultrasonograms, which left 4995 patients in our efficacy population (median treatment duration, 28 days [IQR, 25.0 to 29.0 days] for both groups).

Baseline characteristics were well balanced between groups (Table 1). In both groups and trial periods, most patients had acute infection without septic shock, acute respiratory insufficiency, or heart failure (New York Heart Association class III or IV). The pre- and postamendment safety populations differed in some characteristics, most notably the proportion of patients with level 1 and level 2 immobility (preamendment, 1179 [27.3%] vs. 3135 [72.7%], respectively; postamendment, 1394 [85.7%] vs. 233 [14.3%]) (Figure 1).

# Efficacy

Extended-duration enoxaparin significantly reduced VTE at  $28 \pm 4$  days in the total efficacy population (61 events [2.5%] in the enoxaparin group vs. 100 events [4.0%] in the placebo group; absolute risk difference, -1.53% [95.8% CI, -2.54% to -0.52%]) (Table 2), an effect largely attributable to a decrease in symptomatic DVT

-0.19%]) (Table 3). The effect was unchanged at 90 days, with an additional 4 events in the enoxaparin group and 5 events in the placebo group (absolute risk difference favoring enoxaparin, -1.57% [95.8% CI, -2.61% to -0.53%]), and was similar among the 324 patients with evaluable efficacy data who were also receiving antiplatelet agents (data not shown; *P* value for interaction = 0.32). Cumulative all-cause mortality did not significantly differ between groups at 30, 90, or 180 days (Appendix Table 1, available at www.annals.org). One patient in the placebo group died of a pulmonary embolism. Efficacy outcomes before and after the amendment are shown in Tables 2 and 3 and Appendix Table 1.

(absolute risk difference, -0.60% [95.8% CI, -1.00% to

Sensitivity analyses conducted for the primary efficacy end point suggested these findings were not dependent on the exclusion of unevaluable patients from the efficacy population (**Appendix 3**, available at www.annals.org).

## Safety

The number of major hemorrhages at 30 days was significantly greater in the extended-duration enoxaparin group than the placebo group (25 events [0.8%] vs. 10 [0.3%] events; absolute risk difference, 0.51% [95% CI, 0.12% to 0.89%]), using a 20-g/L reduction in hemoglobin level as a threshold (Table 2). One patient in the

<i>Table 3</i> —Continued		
	Total Population	
Extended-Duration Enoxaparin (n = 2485)	Placebo ( <i>n</i> = 2510)	Absolute Risk Difference (CI), %†
5 (0.2)	24 (1.0)	-0.75 (-1.19 to -0.32)
60 (2.4)	95 (3.8)	-1.37 (-2.37 to -0.37)
5 (0.2)	20 (0.8)	-0.60 (-1.00 to -0.19)
55 (2.2)	75 (3.0)	-0.77 (-1.69 to 0.14)
1 (0.0)	5 (0.2)	-0.16 (-0.34 to 0.04)
0	4 (0.2)	-0.16 (-0.32 to 0.00)
1 (0.1)	1 (0.1)	0.00 (-0.11 to 0.11)
0	1 (0.0)	-0.04 (-0.12 to 0.04)
65 (2.6)	105 (4.2)	-1.57 (-2.61 to -0.53)
8 (0.3)	28 (1.1)	-0.79 (-1.28 to -0.31)
61 (2.5)	98 (3.9)	-1.45 (-2.46 to -0.44)
6 (0.2)	23 (0.9)	-0.67 (-1.11 to -0.24)
55 (2.2)	75 (3.0)	-0.77 (-1.69 to 0.14)
4 (0.2)	7 (0.3)	-0.12 (-0.39 to 0.15)
2 (0.1)	5 (0.2)	-0.12 (-0.33 to 0.09)
2 (0.1)	2 (0.1)	0.00 (-0.16 to 0.16)
1 (0.0)	2 (0.1)	-0.04 (-0.18 to 0.10)

extended-duration enoxaparin group had hemorrhagic transformation of a stroke and died.

When we used the protocol definition of a major bleeding event (30-g/L reduction in hemoglobin level), the absolute difference in major hemorrhages at 30 days was slightly lower (0.44% [95% CI, 0.10% to 0.78%]) (Appendix Table 2, available at www.annals.org). Major hemorrhage rates were similar overall and across trial periods and immobility levels regardless of whether we used a 20-g/L or 30-g/L threshold (Appendix Table 3, available at www.annals.org).

Total bleeding events (major and minor) were also significantly increased in patients who received enoxaparin (absolute risk difference favoring placebo, 2.37% [95% CI, 1.26% to 3.48%]) (Table 4 and Appendix Table 4). A higher proportion of the 411 patients (7%) who received antiplatelet agents for 1 day or longer experienced increases in major bleeding (9 [2.2%] vs. 18 [0.3%]; P < 0.001) and total bleeding events (32 [7.8%] vs. 270 [4.9%]; P < 0.010), but analyses revealed no statistically significant interaction between antiplatelet medication and treatment assignment for either outcome (P = 0.54 for major bleeding and 0.50 for total bleeding).

The proportion of serious adverse events that led to death was 1.3% (39 of 2975 patients) in the extendedduration enoxaparin group and 1.5% (45 of 2988 patients) in the placebo group (**Table 4**). **Table 4** also shows the occurrences of other adverse events. We counted 13 cases of VTE and 20 major bleeding events (hemoglobin level decrease  $\geq 20$  g/L) as serious adverse events (Appendix Table 5, available at www.annals.org). Treatment-emergent adverse events that the investigator judged to be possibly related to study treatment were reported in 1.8% (54 of 2975) and 1.6% (49 of 2988) patients receiving extendedduration enoxaparin and placebo, respectively. The incidence of serious treatment-emergent adverse events that were possibly related to study treatment was 0.2% (5 of 2975 patients) in the extended-duration enoxaparin group and 0.2% (7 of 2988 patients) in the placebo group.

# Subgroup Analyses

Tests of interaction in the total efficacy population revealed a statistically significant difference in primary efficacy outcome by sex (P = 0.016) and age (P = 0.011)(Appendix Figure 1, available at www.annals.org). Sex and age subgroups had distinct characteristics and did not substantially overlap; 35% of the women were older than 75 years, and women represented 59% of patients older than 75 years. We therefore assessed primary efficacy and safety end points by subgroups of sex and age and by immobility level (given the importance of immobility to the study design). Table 5 presents primary efficacy and safety end points for age subgroups. We found statistically significant reductions in VTE with enoxaparin in women (absolute risk difference, -2.71% [95.8% CI, -4.15% to -1.28%]) but not men (-0.36% [95.8% CI, -1.79% to 1.07%]). Similarly, we found statistically significant increases in major bleeding events in women (0.66% [95% CI, 0.11% to (1.21%) but not men (0.34%) [95% CI, -0.20% to 0.89%]) (Appendix Tables 5, 6, and 7, available at www .annals.org). Patients older than 75 years experienced statistically significant reductions in VTE with enoxaparin at all immobility levels (-4.25% [95.8% CI, -6.45% to -2.04%]) and a non-statistically significant increase in major bleeding events (0.24% [95% CI, -0.46% to 0.94%]). Patients 75 years or younger experienced no clear treatment benefit and some harm in both immobility groups (absolute risk difference for increase in major bleeding events, 0.62% [95% CI, 0.15% to 1.08%]).

#### DISCUSSION

Extended-duration prophylaxis with subcutaneous enoxaparin reduced the combined incidence of symptomatic or asymptomatic DVT, symptomatic pulmonary embolism, or fatal pulmonary embolism in acutely ill medical patients with level 1 immobility, those older than 75 years, and women, with increased rates of major bleeding events. The observed absolute differences in risk for VTE and major bleeding events suggest a favorable benefit-to-risk ratio in these patient subgroups, especially those older than 75 years. Our findings do not support the use of extendedduration prophylaxis for patients with level 2 immobility without any of the 3 specified risk factors for VTE. Mortality did not seem to differ between groups at any time point.

An interaction analysis identified patients older than 75 years and women as independent patient subgroups that may particularly benefit from extended-duration prophylaxis (**Table 5** and **Appendix Tables 6** and 7, available at

## Table 4. Incidence of Secondary Safety End Points and Adverse Events

End Point	Preamendment*			Postamendment		
	Extended- Duration Enoxaparin (n = 2159)	Placebo ( <i>n</i> = 2176)	Absolute Risk Difference (95% CI), %	Extended- Duration Enoxaparin (n = 816)	Placebo ( <i>n</i> = 812)	Absolute Risk Difference (95% Cl), %
Bleeding, n (%)						
All patients with bleedingt	157 (7.3)	98 (4.5)	2.77 (1.37 to 4.17)	29 (3.6)	18 (2.2)	1.34 (-0.29 to 2.96)
Patients with major bleeding events‡ Major bleeding events	19 (0.9)	10 (0.5)	0.42 (-0.07 to 0.91)	6 (0.7)	0 (0.0)	0.74 (0.15 to 1.32)
Fatal	1 (<0.1)	0 (0.0)	0.05 (-0.04 to 0.14)	0 (0.0)	0 (0.0)	-
Intracranial	4 (0.2)	0 (0.0)	0.19 (0.00 to 0.37)	0 (0.0)	0 (0.0)	-
Intraocular	0 (0.0)	0 (0.0)	-	1 (0.1)	0 (0.0)	0.12 (-0.12 to 0.36)
Other	16 (0.7)	10 (0.5)	0.28 (-0.18 to 0.74)	5 (0.6)	0 (0.0)	0.61 (0.08 to 1.15)
Minor bleeding events	140 (6.5)	88 (4.0)	2.44 (1.11 to 3.77)	24 (2.9)	18 (2.2)	0.72 (-0.81 to 2.26)
Adverse events, n (%)						
Serious adverse events§	172 (8.0)	170 (7.8)	0.15 (-1.45 to 1.76)	44 (5.4)	48 (5.9)	-0.52 (-2.76 to 1.72)
Treatment-emergent adverse events	602 (27.9)	572 (26.3)	1.60 (-1.05 to 4.24)	146 (17.9)	161 (19.8)	-1.94 (-5.74 to 1.86)
Leading to discontinuation of study drug	82 (3.8)	114 (5.2)	-1.44 (-2.68 to -0.21)	23 (2.8)	23 (2.8)	-0.01 (-1.62 to 1.60)
Leading to death	23 (1.1)	34 (1.6)	-0.50 (-1.17 to 0.18)	16 (2.0)	11 (1.4)	0.61 (-0.63 to 1.85)
Thrombocytopenia (<50 000 $ imes$ 10 <sup>9</sup> cells/L)	6 (0.3)	8 (0.4)	-0.09 (-0.43 to 0.25)	1 (0.1)	1 (0.1)	0.00 (-0.34 to 0.34)
Heparin-induced thrombocytopenia	1 (<0.0)	0	0.05 (-0.04 to 0.14)	0	0	-

VTE = venous thromboembolism.

\* Data include outcomes that were not fully adjudicated at the time of interim analysis. The analysis using adjudicated data only found no significant difference between groups. See "Trial Monitoring and Amendment" in the Methods section for further details.
 † Includes major and minor bleeding events; these data used the post hoc criterion for major bleeding events of a threshold hemoglobin level decrease of 20 g/L. Appendix

† Includes major and minor bleeding events; these data used the post hoc criterion for major bleeding events of a threshold hemoglobin level decrease of 20 g/L. Appendix Table 3 reports incidence of bleeding events with the original threshold criterion for major bleeding events.

\* A threshold decrease in hemoglobin level of 20 g/L was incorporated as a post hoc criterion. Appendix Table 3 reports incidence of major bleeding events with the original threshold criterion of a 30-g/L decrease in hemoglobin level. Appendix Table 7 provides further details about the types of major bleeding events that occurred.

§ Includes 13 VTEs (1 in the enoxaparin group and 12 in the placebo group) and 20 major bleeding events ( $\geq$ 20-g/L decrease in hemoglobin level) (15 in the enoxaparin group and 5 in the placebo group). We also counted VTEs and major bleeding events as serious adverse events if they met the serious adverse event criteria (Methods section and Appendix Table 4).

Adverse events that developed or worsened during the treatment or posttreatment periods and that investigators considered to be possibly related to the study medication. The total includes additional types of events that do not correspond to the listed subcategories.

www.annals.org). PREVENT (Prospective Evaluation of Dalteparin Efficacy for Prevention of VTE in Immobilized Patients Trial) (12) also reported greater reductions in absolute risk for VTE in elderly patients who received lowmolecular-weight heparin, compared with the overall population (3.8% vs. 2.19%, respectively). Previous studies (13) have demonstrated the effect of older age on VTE risk. Differential effects of anticoagulation therapy according to patient sex have also been reported in the context of VTE recurrence when oral anticoagulant therapy is discontinued after an unprovoked VTE; men have a higher risk for recurrence than women (14, 15).

The efficacy of short-term thromboprophylaxis in reducing the risk for VTE in acutely ill medical patients has been reported (2–4) and seems to be driven by a decrease in asymptomatic events. The use of asymptomatic proximal DVT as a clinical efficacy end point in phase 3 studies has been questioned. However, recent data link asymptomatic DVT in atrisk hospitalized medical patients with development of symptomatic VTE, the postthrombotic syndrome, and increased mortality; estimates (6, 16–18) suggest that approximately 34% of VTE-related deaths result from sudden pulmonary embolism in patients with no previous symptomatic VTE diagnosis. Although most confirmed VTEs in our study were asymptomatic proximal DVT diagnosed by ultrasonography, extended-duration prophylaxis was associated with a statistically significant decrease in the incidence of symptomatic VTE compared with placebo, which previous studies of shortterm VTE prophylaxis in acutely ill medical patients did not show (2–4).

We also observed a statistically significant increase in the incidence of bleeding with extended-duration enoxaparin compared with placebo. The rates and absolute differences in major bleeding events observed in patients who received extended-duration enoxaparin are similar to the increased incidence associated with anticoagulant use in placebo-controlled trials of short-term prophylaxis (2–4), and they did not differ when we used a 30-g/L threshold decrease in hemoglobin level instead of a 20-g/L threshold as the criterion for major bleeding.

Our study has limitations. When we designed it, few data were available on the characteristics of acutely ill medical patients who would benefit the most from extendedduration prophylaxis. At the final interim analysis, we observed no significant between-group differences due to lower-than-expected VTE rates in patients with adjudicated data available. The subsequent eligibility criteria amendment focused enrollment on patients with increased VTE risk. This shift in eligibility during the trial means that although we report data for the total trial population,

Total Population					
Extended- Duration Enoxaparin (n = 2975)	Placebo ( <i>n</i> = 2988)	Absolute Risk Difference (95% Cl), %			
186 (6.3)	116 (3.9)	2.37 (1.26 to 3.48)			
25 (0.8)	10 (0.3)	0.51 (0.12 to 0.89)			
1 (<0.1)	0 (0.0)	0.03 (-0.03 to 0.10)			
4 (0.1)	0 (0.0)	0.13 (0.00 to 0.27)			
1 (<0.1)	0 (0.0)	0.03 (-0.03 to 0.10)			
21 (0.7)	10 (0.3)	0.37 (0.01 to 0.74)			
164 (5.5)	106 (3.5)	1.97 (0.91 to 3.02)			
216 (7.3)	218 (7.3)	-0.04 (-1.35 to 1.28)			
748 (25.1)	733 (24.5)	0.61 (-1.58 to 2.80)			
105 (3.5)	137 (4.6)	-1.06 (-2.06 to -0.05)			
39 (1.3)	45 (1.5)	-0.20 (-0.79 to 0.40)			
7 (0.2)	9 (0.3)	-0.07 (-0.33 to 0.20)			
1 (<0.0)	0	0.03 (-0.03 to 0.10)			

Table 4—Continued

our overall findings are no longer generalizable to the whole patient population. In particular, the postamendment population had a notably larger proportion of patients with level 1 immobility than the preamendment population. This may partly have resulted from the continued recruitment of patients with level 1 but not level 2 immobility, pending development and approval of the amendment. We cannot exclude the possibility that some enrolling investigators continued to follow this guidance of level 1-only recruitment after the amendment was issued. In addition, the increased proportion of patients recruited with level 1 immobility may have reflected the relative complexity of patient eligibility assessment, given that enrollment of patients with level 1 immobility did not require screening for additional VTE risk factors. However, despite the amendment's limitations, its implementation and the continuation of the trial allowed us to identify which acutely ill medical patients would and would not benefit from extended-duration prophylaxis.

#### Table 5. Incidence of Primary Efficacy and Safety End Points in Patient Age Subgroups, by Sex and Immobility Level

End Point	Extended-Duration Enoxaparin, n/N (%)	Placebo, n/N (%)	Absolute Risk Difference (CI), %*
VTE at day 28†			
Age >75 y			
Women	9/400 (2.25)	34/446 (7.62)	-5.37 (-8.34 to -2.41)
Level 1	2/179 (1.12)	16/195 (8.21)	-7.09 (-11.39 to -2.78)
Level 2	7/220 (3.18)	18/250 (7.20)	-4.02 (-8.12 to 0.09)
Men	9/325 (2.77)	16/297 (5.39)	-2.62 (-5.86 to 0.63)
Level 1	3/120 (2.50)	8/119 (6.72)	-4.22 (-9.72 to 1.27)
Level 2	6/204 (2.94)	8/178 (4.49)	-1.55 (-5.52 to 2.42)
Age ≤75 y			
Women	14/837 (1.67)	23/801 (2.87)	-1.20 (-2.70 to 0.3)
Level 1	5/372 (1.34)	12/317 (3.79)	-2.44 (-4.94 to 0.05)
Level 2	9/460 (1.96)	10/478 (2.09)	-0.14 (-2.01 to 1.73)
Men	29/923 (3.14)	27/966 (2.80)	0.35 (-1.24 to 1.94)
Level 1	15/399 (3.76)	11/409 (2.69)	1.07 (-1.46 to 3.60)
Level 2	14/521 (2.69)	16/553 (2.89)	-0.21 (-2.25 to 1.84)
Major bleeding events‡			
Age >75 y			
Women	4/493 (0.8)	3/549 (0.5)	0.26 (-0.74 to 1.27)
Level 1	2/227 (0.9)	2/250 (0.8)	0.08 (-1.56 to 1.72)
Level 2	2/265 (0.8)	1/298 (0.3)	0.42 (-0.81 to 1.65)
Men	2/385 (0.5)	1/354 (0.3)	0.24 (-0.67 to 1.14)
Level 1	2/149 (1.3)	0/151 (0.0)	1.34 (-0.51 to 3.19)
Level 2	0/235 (0.0)	1/203 (0.5)	-0.49 (-1.46 to 0.47)
Age ≤75 y			
Women	10/1015 (1.0)	1/962 (0.1)	0.88 (0.24 to 1.52)
Level 1	3/446 (0.7)	0/406 (0.0)	0.67 (-0.09 to 1.43)
Level 2	7/564 (1.2)	1/550 (0.2)	1.06 (0.08 to 2.04)
Men	9/1082 (0.8)	5/1123 (0.4)	0.39 (-0.28 to 1.05)
Level 1	2/470 (0.4)	0/474 (0.0)	0.43 (-0.16 to 1.01)
Level 2	7/608 (1.2)	5/645 (0.8)	0.38 (-0.71 to 1.46)

VTE = venous thromboembolism.

\* For VTE end points in the total population, we report 95.8% CIs (P < 0.042) because of the  $\alpha$  adjustment for the interim analysis. For all other end points, we report 95% CIs (P < 0.050).

+ Assessed in the efficacy population (4995 patients). We excluded 21 patients from the immobility subgroups because of missing immobility-level data or absence of immobility-level classification. Of these, 1 had VTE (a woman >75 years who received placebo).

\* A threshold decrease in hemoglobin level of 20 g/L was incorporated as a post hoc criterion. Appendix Table 6 reports incidence of major bleeding events with the original threshold criterion of a 30-g/L decrease in hemoglobin level. Assessed in the safety population (5963 patients). We excluded 22 patients from this analysis because of missing immobility-level data or absence of immobility-level classification; we observed no major bleeding events among these patients.

# ARTICLE | Extended-Duration Venous Thromboembolism Prophylaxis in Medical Patients

Another limitation is that we do not report the number of patients screened for inclusion. Although these data are important for assessing the generalizability of the observed results, we did not systematically capture them during our study. Finally, although our findings suggest that extended-duration VTE prophylaxis has a favorable balance of benefits and harms (17), decisions about the use of prophylaxis should depend on individual patient characteristics, preferences, and values (19, 20).

In conclusion, compared with placebo, extendedduration enoxaparin prophylaxis was associated with a reduction in the combined incidence of symptomatic or asymptomatic DVT, symptomatic pulmonary embolism, and fatal pulmonary embolism in acutely ill medical patients with level 1 immobility, those older than 75 years, and women. This risk reduction occurred in addition to that associated with short-term prophylaxis and was associated with levels of major bleeding events similar to those seen in previous studies of short-term VTE prophylaxis in medical patients (2–4). These findings, in addition to a patient's individual benefit–risk assessment and preferences, should assist physicians who are considering the use of extended-duration prophylaxis.

From University of Calgary, Foothills Hospital, Calgary, Alberta, and McMaster University and HHSC McMaster Clinic, Hamilton, Ontario, Canada; Hospital Carl Gustav Carus, Dresden, Germany; Duke University Medical Center, Durham, North Carolina; Hospital Germans Trias i Pujol, Barcelona, Spain; Service d'Hématologie Biologique Hotel-Dieu and sanofi-aventis, Paris, France; and Washington University School of Medicine, St. Louis, Missouri.

Acknowledgment: The authors thank Min Chen and Genevieve Salette (sanofi-aventis) for their statistical analysis, Rachel Spice (Excerpta Medica) for providing editorial support, and Bruno Deslandes (sanofi-aventis) for critically reviewing the content.

Grant Support: By sanofi-aventis, which also funded editorial support.

**Potential Conflicts of Interest:** Disclosures can be viewed at www.acponline .org/authors/icmje/ConflictOfInterestForms.do?msNum=M09-0393.

**Reproducible Research Statement:** *Study protocol:* See reference 11. *Statistical code:* Not available. *Data set:* Available on request from the EXCLAIM Steering Committee (e-mail, rdhull@ucalgary.ca).

Requests for Single Reprints: Russell D. Hull, MBBS, Thrombosis Research Unit, University of Calgary, Foothills Hospital, Room 601 South Tower, 1403 29th Street Northwest, Calgary, Alberta T2N 2T9, Canada; e-mail, rdhull@ucalgary.ca.

Current author addresses and author contributions are available at www .annals.org.

# References

 Geerts WH, Bergqvist D, Pineo GF, Heit JA, Samama CM, Lassen MR, et al; American College of Chest Physicians. Prevention of venous thromboembolism: American College of Chest Physicians Evidence-Based Clinical Practice Guidelines (8th Edition). Chest. 2008;133:381S-453S. [PMID: 18574271]
 Samama MM, Cohen AT, Darmon JY, Desjardins L, Eldor A, Janbon C,

et al. A comparison of enoxaparin with placebo for the prevention of venous

thromboembolism in acutely ill medical patients. Prophylaxis in Medical Patients with Enoxaparin Study Group. N Engl J Med. 1999;341:793-800. [PMID: 10477777]

3. Leizorovicz A, Cohen AT, Turpie AG, Olsson CG, Vaitkus PT, Goldhaber SZ; PREVENT Medical Thromboprophylaxis Study Group. Randomized, placebo-controlled trial of dalteparin for the prevention of venous thromboembolism in acutely ill medical patients. Circulation. 2004;110:874-9. [PMID: 15289368]

4. Cohen AT, Davidson BL, Gallus AS, Lassen MR, Prins MH, Tomkowski W, et al; ARTEMIS Investigators. Efficacy and safety of fondaparinux for the prevention of venous thromboembolism in older acute medical patients: randomised placebo controlled trial. BMJ. 2006;332:325-9. [PMID: 16439370]

5. Baglin TP, White K, Charles A. Fatal pulmonary embolism in hospitalised medical patients. J Clin Pathol. 1997;50:609-10. [PMID: 9306945]

6. Cohen AT, Agnelli G, Anderson FA, Arcelus JI, Bergqvist D, Brecht JG, et al; VTE Impact Assessment Group in Europe (VITAE). Venous thromboembolism (VTE) in Europe. The number of VTE events and associated morbidity and mortality. Thromb Haemost. 2007;98:756-64. [PMID: 17938798]

7. Hull RD, Pineo GF, Stein PD, Mah AF, MacIsaac SM, Dahl OE, et al. Extended out-of-hospital low-molecular-weight heparin prophylaxis against deep venous thrombosis in patients after elective hip arthroplasty: a systematic review. Ann Intern Med. 2001;135:858-69. [PMID: 11712876]

8. Eikelboom JW, Quinlan DJ, Douketis JD. Extended-duration prophylaxis against venous thromboembolism after total hip or knee replacement: a metaanalysis of the randomised trials. Lancet. 2001;358:9-15. [PMID: 11454370]

 Bergqvist D, Agnelli G, Cohen AT, Eldor A, Nilsson PE, Le Moigne-Amrani A, et al; ENOXACAN II Investigators. Duration of prophylaxis against venous thromboembolism with enoxaparin after surgery for cancer. N Engl J Med. 2002; 346:975-80. [PMID: 11919306]

10. Rasmussen MS, Jorgensen LN, Wille-Jørgensen P, Nielsen JD, Horn A, Mohn AC, et al; FAME Investigators. Prolonged prophylaxis with dalteparin to prevent late thromboembolic complications in patients undergoing major abdominal surgery: a multicenter randomized open-label study. J Thromb Haemost. 2006;4:2384-90. [PMID: 16881934]

11. Hull RD, Schellong SM, Tapson VF, Monreal M, Samama MM, Turpie AG, et al. Extended-duration thromboprophylaxis in acutely ill medical patients with recent reduced mobility: methodology for the EXCLAIM study. J Thromb Thrombolysis. 2006;22:31-8. [PMID: 16786230]

12. Kucher N, Leizorovicz A, Vaitkus PT, Cohen AT, Turpie AG, Olsson CG, et al. Efficacy and safety of fixed low-dose dalteparin in preventing venous thromboembolism among obese or elderly hospitalized patients: a subgroup analysis of the PREVENT trial. Arch Intern Med. 2005;165:341-5. [PMID: 15710801]

13. Stein PD, Hull RD, Kayali F, Ghali WA, Alshab AK, Olson RE. Venous thromboembolism according to age: the impact of an aging population. Arch Intern Med. 2004;164:2260-5. [PMID: 15534164]

14. Rodger MA, Kahn SR, Wells PS, Anderson DA, Chagnon I, Le Gal G, et al. Identifying unprovoked thromboembolism patients at low risk for recurrence who can discontinue anticoagulant therapy. CMAJ. 2008;179:417-26. [PMID: 18725614]

15. McRae S, Tran H, Schulman S, Ginsberg J, Kearon C. Effect of patient's sex on risk of recurrent venous thromboembolism: a meta-analysis. Lancet. 2006; 368:371-8. [PMID: 16876665]

 Wille-Jørgensen P, Jorgensen LN, Crawford M. Asymptomatic postoperative deep vein thrombosis and the development of postthrombotic syndrome. A systematic review and meta-analysis. Thromb Haemost. 2005;93:236-41. [PMID: 15711738]

17. Lloyd NS, Douketis JD, Moinuddin I, Lim W, Crowther MA. Anticoagulant prophylaxis to prevent asymptomatic deep vein thrombosis in hospitalized medical patients: a systematic review and meta-analysis. J Thromb Haemost. 2008;6:405-14. [PMID: 18031292]

18. Vaitkus PT, Leizorovicz A, Cohen AT, Turpie AG, Olsson CG, Goldhaber SZ; PREVENT Medical Thromboprophylaxis Study Group. Mortality rates and risk factors for asymptomatic deep vein thrombosis in medical patients. Thromb Haemost. 2005;93:76-9. [PMID: 15630494]

19. Hirsh J, Guyatt G, Lewis SZ. Reflecting on eight editions of the American College of Chest Physicians antithrombotic guidelines [Editorial]. Chest. 2008; 133:1293-5. [PMID: 18574282]

20. Hull RD. Thromboprophylaxis in knee arthroscopy patients: revisiting values and preferences [Editorial]. Ann Intern Med. 2008;149:137-9. [PMID: 18626053]

# **Annals of Internal Medicine**

**Current Author Addresses:** Dr. Hull: Thrombosis Research Unit, University of Calgary, Foothills Hospital, Room 601 South Tower, 1403 29th Street Northwest, Calgary, Alberta T2N 2T9, Canada.

Dr. Schellong: Division of Angiology, University Hospital Carl Gustav Carus, Fetscherstraße 74, 01307 Dresden, Germany.

Dr. Tapson: Duke University Medical Center, Hanes House, Room 128, Durham, NC 27710.

Dr. Monreal: Carretera de Canyet s/n, Barcelona, 08916, Spain.

Dr. Samama: Départment d'Hématologie Biologique, Hôtel Dieu, 75181 Paris, France.

Dr. Nicol: sanofi-aventis, 174 Avenue de France, 75635 Paris Cedex 13, France.

Dr. Vicaut: Clinical Research Unit, Hôpital F. Widal, 200 Rue du Faubourg St-Denis, 75010 Paris, France.

Dr. Turpie: Hamilton General Hospital, 237 Barton Street East, Hamilton, Ontario L8I 2X2, Canada.

Dr. Yusen: Washington University School of Medicine, 660 South Euclid Avenue, Box 8052, St. Louis, MO 63110.

Author Contributions: Conception and design: R.D. Hull, S.M. Schellong, V.F. Tapson, M. Monreal, A.G.G. Turpie, R.D. Yusen.

Analysis and interpretation of the data: R.D. Hull, S.M. Schellong, V.F. Tapson, M.M. Samama, E. Vicaut, A.G.G. Turpie, R.D. Yusen.

Drafting of the article: R.D. Hull, M. Monreal, R.D. Yusen.

Critical revision of the article for important intellectual content: R.D. Hull, S.M. Schellong, V.F. Tapson, M. Monreal, M.M. Samama, A.G.G. Turpie, R.D. Yusen.

Final approval of the article: R.D. Hull, S.M. Schellong, V.F. Tapson, M. Monreal, M.M. Samama, A.G.G. Turpie, R.D. Yusen.

Provision of study materials or patients: R.D. Hull, S.M. Schellong, V.F. Tapson, M. Monreal, A.G.G. Turpie, R.D. Yusen.

Statistical expertise: R.D. Hull, S.M. Schellong, E. Vicaut, R.D. Yusen. Administrative, technical, or logistic support: S.M. Schellong, P. Nicol. Collection and assembly of data: R.D. Hull, V.F. Tapson, M. Monreal, M.M. Samama, A.G.G. Turpie, P. Nicol, R.D. Yusen.

# APPENDIX 1: THE EXCLAIM STUDY INVESTIGATORS Steering Committee

Russell D. Hull, Manuel Monreal, Meyer-Michel Samama, Sebastian M. Schellong, Victor F. Tapson, Alexander G.G. Turpie, and Roger D. Yusen.

#### Data Monitoring Committee

Donald Easton, Sylvia Haas, Victor J. Marder, Guy Meyer, and Yuchung J. Wang.

# Ultrasound Scan Adjudication Committee

Sebastian M. Schellong and Kai Halbritter.

#### **Clinical Adjudication Committee**

Russell D. Hull, Manuel Monreal, Meyer-Michel Samama, Sebastian M. Schellong, Victor F. Tapson, Alexander G.G. Turpie, and Roger D. Yusen.

### Key sanofi-aventis Personnel

Philippe Nicol (Study Manager), Catherine Domenger (Medical Adviser), Min Chen (Lead Statistician), Geneviève Salette (Statistician), and Bruno Deslandes (Clinical Director).

#### Enrolling Investigators

Argentina: J.A. Mazzei, M. Casey, M. del Carmen Gallo, J. Ceresetto, and B. Grand.

*Australia:* H. Salem, C. Denaro, V. Ayyar, T. Brighton†, K. Narayan, H. Gibbs, E. Gan, A. Gallus, S. Dunkley, B. Chong, R. Baker, M. Leahy, U. Hahn, P. Blombery, C. Ward, B. McGrath, I. Prosser, and S. Chunilal.

Austria: J. Patsch, M. Jud, K. Aigner, F. Hoppichler, H. Pall, P. Grabher, P. Kobierski, H. Zwick, M. Juchum, M.G.

Röger, H.R. Schönherr, F. Schmalzl, C. Geyer, and B. Kohler. *Belgium:* M. Delcroix, M. Vandewoude, J.C. Wautrecht<sup>†</sup>,K. De Boeck, C. Hermans, and P. Hainaut.

*Brazil:* G. Reis, A.C.P. Barreto, M. de Arruda Martins, C. Carvalho, V. Tadini, J.A. de Barros, L. Moura, A.C. Lopes, C. Pereira da Cunha, C. Gun, and R. Milani.

*Canada:* S. Kahn, M.J. Miron, C. Demers, L. Desjardins, K. Lai, S. Solymoss, J.D. Rolf, R. Audet, A.G.G. Turpie, T. Wong, I. Kirouac, P. Duffy, S. Dolan, J. Kassis, D. Anderson, A. Kuchtaruk, J. Berlingieri, G. Gamble, K. Grewal, M. Weigel, J. Biem, R. Labonte, A. Cheung, A. Forster, A. Panju, J. Villeneuve, and B. McCarron.

*Colombia:* P.M. Pacheco, F. Cuervo, R. Dennis, A. Londoño, E.J. Ruiz, J.C. Velásquez<sup>†</sup>, R. Acero, R.D. Vargas, and U. Largo.

*France:* D. Mottier<sup>†</sup>, A. Proust, I. Quere, H. Decousus, B. Lorcerie, J.F. Bergmann, J.J. Leduc, C. Series, G. Dien, E. Duhamel, G. Berrut, T. Prazuck, M. Goralski, J.Y. Hatron, D. Farge, P. Jego, A. Achkar, P. Lacroix, J. Doucet, O. Bouchard, F. Parent, H. Boccalon, C. Conri, E. Le Moigne, P. Letellier, O. Sanchez, and P. Zuck.

Germany: J. Harenberg, M. Sternkopf, H. Lawall, C. Kelbel, G. Pöhlmann, T. Dorsel, O. Altmann, W. Sehnert, G. Vossbeck, F. Odemar, W. Oettler, A. Hartmann, K. Nogai, H. Bechtold, R. Fünfstück, H. Omran, T. Ittel, R. Egger, T. Horacek, J. Cailloud, S. Maier, G. Hasenfuss, A. Schmidt-Lucke, M. Pfisterer, W. Lengfelder, J. Treib, H.J. Dieckmann, H. Thomas, T. Ziegler, J. Brachmann, F.L. Dumoulin, G. Cieslinski, B. Schmidt, and S. Schellong<sup>†</sup>.

*India:* P. Mehta, K. Asokan, P.M. Sathe, R.M.P.L. Ramanathan, M.V.S. Subbalaxmi, Z. Udwadia, S.K. Sharma, N. Ramakrishnan, and A. Vigg.

Israel: A. Shlomo Berliner, G. Lugassy, M. Ellis, and B. Brenner.

*Italy:* E. De Gaudenzi, D. Imberti, M. Silingardi, S. Siragusa, C. Cimminiello<sup>†</sup>, P. Prandoni, W. Ageno, G. Scannapieco, N. Zanatta, M. Pini, S. Testa, L. Lusiani, P. Pola, G. Marchegiani, and G.M. Patrassi.

*Mexico:* D. Rodriguez-Gonzalez, U. Chavarria-Martinez, J.F. Velasco-Rodriguez, J. Jaimes-Hernandez, A. Cruz-Diaz, F. Aguilera-Almazan, A. Arauz Góngora†, M. Poblano-Morales, G. Velasco-Sanchez, A. Quesada-Sanchez, R. Lara-Badillo, A. Mireles-Munoz, and V. Borja-Sanchez.

*Poland:* A. Torbicki, P. Psuja, P. Pruszczyk†, J. Musial, G. Opolski, M. Dluzniewski, and K. Zawilska.

*Russia:* V.A. Lyusov, I. Gordeev, P.Y. Dovgalevskiy, S.B. Fitilev, M.G. Glezer, G.I. Storozhakov, M.A. Karpenko, K.E. Sobolev, O.A. Khrustalev, A.G. Chuchalin, A.L. Vertkin, S.A. Rumyantseva, U.B. Belousov, D.A. Zateytshikov, and M.A. Piradov.

W-2 6 July 2010 Annals of Internal Medicine Volume 153 • Number 1

South Africa: J. Verster, J. Engelbrecht, F. Maritz, L. van Zyl, G.H. Latiff, and M.M. De Vries Basson<sup>†</sup>.

Spain: M. Monreal<sup>†</sup>, J. Luis Rodriguez, D. Jimenez, J. Moreno, J. Ramon Calabuig, C. Suarez, A. Castro, S. Reus, C. Falga, R. Otero, M. Aburto, P. Conthe, A. Page, A. Blanco, A. Valencia, F. Martos, J. Ruiz Galiana, and J. Trujillo.

*Tunisia:* F. Boujelben, R. Hassine, F. Bahri, A. Derbel, S. Ben Youssef Zouari, S. Othmani, R. Charfi, M. Chakroun, H. Bouacha, M. Marouane, S. Mahjoub, R. Miladi, C. Laouani Kechrid, N. Belkhiria, H. Drissa, and H. Houman<sup>†</sup>.

United Kingdom: A. Cohen<sup>†</sup>, M. Gormley, and A.J. Moriarty. United States: W. Rodriguez, J. Updegrove, M. Hazelrigg, S. Etezadi, D. Amin, K. Krell, V. Nadar, R. Yusen<sup>†</sup>, J. Dexter, R. Fei, A. Marinelli, A. Seibert, J. Welker, W. O'Riordan, E. Kolettis, D. Kett, D. Hill, D. Buchan, T. Morris, M. Siegel, A. Spyropoulos, R. Lerner, W. Leeds, M. Anderson III, D. Banish, J. Anderson, F. Messina, A. Baker, L. Emdur, V. Robinson, K. Danisa, B. Sangani, D. Lawler, R. Betzu, L. Gatien, H. Berlin, R. Deichmann, S. Kaatz, M. Amin, H. Fleming, W. McKenzie, C. Lawton, A.J. Quaranta, A. Shah, M. Rush, R. Snyder, R. Shriver, J. Ansell, V. Tapson, R.C. Touchon, R.T.P. Chow, H. Thawani, S.J. Simon, S. Deitelzweig, R. McLafferty, S. Jacobson, B. Davidson, R. Elliott-Mullens, P. Patel, M. Waters, J. Hansbrough, N. Abramson, B. Lahiri, M. El Shahawy, R. Levine, J. Fenton, W.J. Pendergast, R. Lavender, K.L. Watson-Ramirez, C. Pollack, B. Chamacho, T. Albertson, D. Beard, J.R. Rehem, D. Elias, M. Pistoria, H. Chandarana, E. Elamin, W. Miller, C.L. Anderson, K. Leeper, A. Schlau, C. Sotolongo, H. Karunartne, G.L. Walters, W. Summer, L.C. Faulk, L. Barr, L. Thet, R.D. Hite, D. Ost, G. Criner, G. Martin, P. Kaboli, D. Brofman, J. Johnson, L. Rink, G. Merli, L. Kendrick, G. Schuyler, D. Chardon, M.C. Joseph, C. Francis, C. Lewinstein, S. Mauger, A. El-Solh, H. Yu, D. Green, S. Padove, J. Sippel, P. Norwood, H. Haught, C. Palmer, M. Moncure, P.W. Sturm, A.K. Gupta, T. Lee, R. Ali, N. Desbiens, J. Witt, L. Greenspon, and D. Heiselman.

† National coordinator.

## APPENDIX 2: SUPPLEMENTARY METHODOLOGY

#### Central Adjudication of Efficacy and Safety End Points

A subcommittee of the steering committee performed adjudication of hemorrhages (presence or absence and severity), deaths (death characteristics and relation with treatment), and suspected VTE (including diagnostic tests, such as chest computed tomography, pulmonary angiography, venography, and ventilation–perfusion lung scan). The adjudication committees that assessed the efficacy and safety end points were blinded to treatment allocation but were aware of the period of the study when events occurred: open-phase, double-blind, or follow-up. A quorum of at least 3 members, including the chairperson, attended each adjudication meeting.

# Event Rate Assumptions and Modified Target Enrollment After the Protocol Amendment

After the eligibility criteria were amended, we assumed that the placebo group in the higher-risk population would have a VTE rate of 5.2% at day 28 after random assignment. Given this increase in expected VTE rate, a type I error rate of 4.7% (adjusted following the interim analysis), and 80% power, we needed a reduced total of 3072 evaluable patients (1536 per treatment group) who met amendment criteria to show a 40% reduction in VTE incidence with the extended-duration enoxaparin regimen. Assuming an evaluability rate of 65%, we aimed to enroll 4730 patients who met the amendment criteria. Because we had already recruited 2883 patients with either level 1 or level 2 immobility (who were older than 75 years or had cancer, history of cancer, or history of VTE) at the time of the interim analysis and eligibility criteria amendment, we only needed an additional 1843 patients to achieve the revised target number of evaluable patients.

# APPENDIX 3: SENSITIVITY ANALYSES CONDUCTED FOR THE PRIMARY EFFICACY END POINT

We conducted sensitivity analyses to assess whether inclusion of all randomly assigned patients excluded from the efficacy population changed the primary efficacy end point result, assuming observed within-group event rates for missing patients and using the upper boundary of VTE incidence in the placebo group to determine the VTE incidence in the enoxaparin group required to yield a statistically nonsignificant difference. The difference in primary outcome remained statistically significant in both analyses, which suggests that the finding did not depend on the exclusion of the unevaluable patients.

Approximately 18% (1090 of 6085) of all randomly assigned patients were unevaluable and were therefore excluded from the primary efficacy analysis (including 122 patients who were randomly assigned but did not receive any prophylaxis during the double-blind phase and 968 patients who were randomly assigned, received treatment, but were excluded from the efficacy population for different reasons) (**Figure 1**). The unevaluable patients comprised 549 patients in the extended-duration enoxaparin group and 541 patients in the placebo group. Because we did not conduct the primary efficacy analysis on a true intentionto-treat basis, we subsequently performed a post hoc sensitivity analysis on all randomly assigned patients.

We performed 2 separate sensitivity analyses on the basis of 2 different key assumptions. The first analysis assumed that a similar rate of VTE would be observed in both patients who were included in the primary analysis of efficacy and those who were excluded (2.5% in the extended-duration enoxaparin group and 4.0% in the placebo group). On the basis of this assumption, we predicted a VTE rate of 76 of 3034 in the extended-duration enoxaparin group and 122 of 3051 in the placebo group, with a level of significance of P < 0.001.

In the second analysis, we assumed that the incidence rate of VTE in patients excluded from the primary efficacy analysis (549 and 541 patients in the enoxaparin and placebo groups, respectively) would range from the lower boundary (3%) to the upper boundary (5%) of the 95% CI in the placebo group and from 2% to 10% in the enoxaparin group, although the upper boundary of the 95% CI of observed VTE incidence in the enoxaparin group was only 3.1%. Assuming a VTE incidence rate of 5% in

the 541 patients in the placebo group (worst case), the VTE rate among the 549 patients in the enoxaparin group should be greater than 6.4% to obtain a nonsignificant difference between the groups (P > 0.042). Because the patients assessed for primary efficacy and those who were unevaluable (and therefore excluded from this assessment) were demographically similar, the likelihood of observing such results can be considered low.

Appendix Figure 2 shows changes in the P value according to the rates of VTE in the placebo group.

In conclusion, the strong significance of the treatment effect observed in the primary efficacy analysis would not have been affected by the inclusion of unevaluable patients.

# Appendix Table 1. All-Cause Mortality, by Immobility Level\*

Time Point and Immobility Level	All-Cause Mortality			
· · · · ·	Extended-Duration Enoxaparin, n/N (%)	Placebo, n/N (%)	Hazard Ratio (95% CI)	
Day 30				
Total	60/2975 (2.1)	65/2988 (2.2)	0.93 (0.65 to 1.32)	
Preamendment	43/2159 (2.1)	47/2176 (2.2)	0.93 (0.61 to 1.40)	
Postamendment Immobility level	17/816 (2.1)	18/812 (2.3)	0.93 (0.48 to 1.80)	
Level 1	36/1292 (2.8)	43/1281 (3.4)	0.83 (0.53 to 1.29)	
High-risk level 2†	18/721 (2.6)	11/746 (1.5)	1.70 (0.80 to 3.59)	
Low-risk level 2‡	6/951 (0.7)	11/950 (1.2)	0.55 (0.20 to 1.48)	
Day 90				
Total	148/2975 (5.2)	142/2988 (5.0)	1.04 (0.83 to 1.31)	
Preamendment	105/2159 (5.2)	105/2176 (5.1)	1.01 (0.77 to 1.32)	
Postamendment Immobility level	43/816 (5.4)	37/812 (4.7)	1.14 (0.74 to 1.77)	
Level 1	82/1292 (6.6)	76/1281 (6.1)	1.06 (0.78 to 1.45	
High-risk level 2†	43/721 (6.3)	41/746 (5.9)	1.08 (0.71 to 1.66)	
Low-risk level 2‡	23/951 (2.6)	25/950 (2.8)	0.92 (0.52 to 1.61)	
Day 180				
Total	220/2975 (8.2)	204/2988 (7.7)	1.08 (0.89 to 1.31)	
Preamendment	153/2159 (8.1)	150/2176 (7.8)	1.03 (0.82 to 1.29	
Postamendment Immobility level	67/816 (8.5)	54/812 (7.5)	1.23 (0.86 to 1.75)	
Level 1	123/1292 (10.1)	105/1281 (9.2)	1.16 (0.89 to 1.50)	
High-risk level 2†	63/721 (10.2)	57/746 (8.4)	1.15 (0.80 to 1.64)	
Low-risk level 2‡	34/951 (4.1)	42/950 (5.3)	0.80 (0.51 to 1.26	

VTE = venous thromboembolism.

\* We assessed mortality in the safety population and used Kaplan–Meier estimates to determine all-cause mortality rates.

† Patients with level 2 immobility and ≥1 of the following VTE risk factors: age 75 y, history of VTE, or active or previous cancer.

**‡** Patients with level 2 immobility and none of the additional specified risk factors for VTE.

-		Preamendmen	±		Postamendm	lent		Total Populatic	Ę
	Extended- Duration Enoxaparin, n/N (%)	Placebo, n/N (%)	Absolute Risk Difference (95% Cl), %	Extended- Duration Enoxaparin, n/N (%)	Placebo, n/N (%)	Absolute Risk Difference (95 % Cl), %	Extended- Duration Enoxaparin, n/N (%)	Placebo, n/N (%)	Absolute Risk Difference (95% CI), %
	16/2159 (0.7)	7/2176 (0.3)	0.42 (-0.01 to 0.85)	4/816 (0.5)	0/812 (0.0)	0.49 (0.01 to 0.97)	20/2975 (0.7)	7/2988 (0.2)	0.44 (0.10 to 0.78)
1	4/590 (0.7)	0/589 (0.0)	0.68 (0.02 to 1.34)	2/702 (0.3)	0/692 (0.0)	0.28 (-0.11 to 0.68)	6/1292 (0.5)	0/1281 (0.0)	0.46 (0.09 to 0.84)
12	12/1559 (0.8)	7/1576 (0.4)	0.33 (-0.22 to 0.87)	2/113 (1.8)	0/120 (0.0)	1.77 (-0.66 to 4.20)	14/1672 (0.8)	7/1696 (0.4)	0.42 (-0.11 to 0.96
gh-risk‡	4/627 (0.6)	3/640 (0.5)	0.17 (-0.65 to 0.99)	2/94 (2.1)	0/106 (0.0)	2.13 (-0.79 to 5.04)	6/721 (0.8)	3/746 (0.4)	0.43 (-0.37 to 1.23
w-risk§	8/932 (0.9)	4/936 (0.4)	0.43 (-0.29 to 1.16)	0/19 (0.0)	0/14 (0.0)	0.00 (0.00 to 0.00)	8/951 (0.8)	4/950 (0.4)	0.42 (-0.29 to 1.13

g/L

Appendix Table 2. Incidence of the Primary Safety End Point, Using the Protocol-Defined Hemoglobin Decrease Threshold of 30

ď'n 200 Assessed in the safety population (2005 patients). We excluded 22 patients in the total safety population (21 preamendment [10 from the enoxaparin group and 11 from the place enoxaparin group]) because of missing immobility-level data or absence of immobility-level classification. † Data include outcomes that were not fully adjudicated at the time of interim analysis. The analysis using adjudicated data only found no significant difference between groups. See

"Trial Monitoring and Amendment" in the Methods section for further details.

following VTE risk factors: age >75 y, history of VTE, or active or previous cancer e additional specified risk factors for VTE.  $| \ge 1$  of the fol 1 none of the zPatients with level 2 immobility and Patients with level 2 immobility and



# Appendix Figure 1. Relative risk for VTE across specific patient subgroups during the double-blind treatment period.

Assessed in the total efficacy population. NYHA = New York Heart Association; OR = odds ratio; VTE = venous thromboembolism.

# Appendix Table 3. Causes of Major Bleeding Events

Major Bleeding Event	Preamendmen	t, n	Postamendme	nt, <i>n</i>	Total Populatio	on, <i>n</i>
	Extended-Duration Enoxaparin*	Placebo	Extended-Duration Enoxaparin	Placebo	Extended-Duration Enoxaparin*	Placebo
Hemoglobin decrease ≥20 g/L†						
Gastrointestinal	9	7	1		10	7
Intracranial	4				4	
Fatal‡	1				1	
Hemothorax	1				1	
Pulmonary hemorrhage	1				1	
Subcutaneous	2				2	
Epistaxis	1				1	
Hematoma		2	1		1	2
Ecchymosis	1				1	
Leg hemorrhage, intermittent epistaxis, and intermittent abdominal ecchymosis			1		1	
Macroscopic hematuria or bladder-wall hematoma	1				1	
Hemoglobin decrease ≥30 g/L						
Gastrointestinal	8	7	1		9	7
Intracranial	4				4	
Fatal‡	1				1	
Hemothorax	1		1		2	
Pulmonary hemorrhage	1				1	
Subcutaneous	2				2	
Epistaxis						
Ecchymosis						
Macroscopic hematuria or bladder-wall hematoma	1				1	
Intraocular					1	
Hemoptysis					1	

\* Two patients included in the extended-duration group of both the preamendment and total populations experienced 2 major bleeding events. One patient had a fatal intracranial bleeding event, and 1 patient experienced both an intracranial and a gastrointestinal bleeding event.

† Includes 1 patient in the placebo group with a hemoglobin decrease of  $\geq 20$  g/L at an undefined location.

**‡** Hemorrhagic transformation of a stroke.



Appendix Figure 2. Changes in the P value, by VTE incidence in the placebo group.

The area below the horizontal dashed line represents the region of statistical significance. VTE = venous thromboembolism.

		Preamendme	ent		Postamendi	nent		Total Populat	ion
	Extended- Duration Enoxaparin (n = 2159), n (%)	Placebo ( <i>n</i> = 2176), <i>n</i> (%)	Absolute Risk Difference (95 % Cl), %	Extended- Duration Enoxaparin ( <i>n</i> = 816), <i>n</i> (%)	Placebo (n = 812), n (%)	Absolute Risk Difference (95% CI), %	Extended- Duration Enoxaparin ( <i>n</i> = 2975), <i>n</i> (%)	Рlacebo (n = 2988), n (%)	Absolute Risk Difference (95% CI), %
All patients with bleeding*	157 (7.3)	98 (4.5)	2.77 (1.37 to 4.17)	29 (3.6)	18 (2.2)	1.34 (-0.29 to 2.96)	186 (6.3)	116 (3.9)	2.37 (1.26 to 3.48)
Patients with major bleeding events† Major bleeding events	16 (0.7)	7 (0.3)	0.42 (-0.01 to 0.85)	4 (0.5)	0 (0.0)	0.49 (0.01 to 0.97)	20 (0.7)	7 (0.2)	0.44 (0.10 to 0.78)
Fatal	1 (<0.1)	0 (0.0)	0.05 (-0.04 to 0.14)	0 (0.0)	0 (0.0)	I	1 (<0.1)	0 (0.0)	0.03 (-0.03 to 0.10)
Intracranial	4 (0.2)	0 (0.0)	0.19 (0.00 to 0.37)	0 (0.0)	0 (0.0)	1	4 (0.1)	0 (0.0)	0.13 (0.00 to 0.27)
Intraocular	0 (0.0)	0 (0.0)	I	1 (0.1)	0 (0.0)	0.12 (-0.12 to 0.36)	1 (<0.1)	0 (0.0)	0.03 (-0.03 to 0.10)
Other	13 (0.6)	7 (0.3)	0.28 (-0.12 to 0.68)	3 (0.4)	0 (0.0)	0.37 (-0.05 to 0.78)	16 (0.5)	7 (0.2)	0.30 (-0.01 to 0.62)
Minor bleeding events	143 (6.6)	91 (4.2)	2.44 (1.10 to 3.79)	26 (3.2)	18 (2.2)	0.97 (-0.60 to 2.54)	169 (5.7)	109 (3.6)	2.03 (0.96 to 3.10)
* The total includes both mi	ajor and minor blee	ding events. Major t	bleeding event criteria include	d the original three	shold criterion of	a 30-g/L decrease in hemoglo	oin level. T <b>able</b> 4 re	ports incidence of n	najor bleeding events with

the post hoc threshold criterion of a 20-g/L decrease in hemoglobin level. Appendix Table 3 provides details about the types of major bleeding events that occurred.

Appendix Table 5. Primary Efficacy and	Safety Outcomes Also	Counted as SAEs				
Outcome	Exte	ended-Duration Enoxaparin, <i>n</i>			Placebo, n	
	Patients	Counted as SAEs	Total SAEs	Patients	Counted as SAEs	Total SAEs
VTE	61	<b>~</b>	216	100	12	218
Bleeding Hemoglobin decrease ≥20 g/L	20	15		7	5	
Hemoglobin decrease ≥30 g/L	25	15		10	7	
AEs = serious adverse events; VTE = venous thro:	mboembolism.					

Appendix Table 4. Incidence of Major and Minor Bleeding Events

## Appendix Table 6. Incidence of Primary Efficacy and Safety End Points, by Sex, Age, and Immobility Level

End Point	Extended- Duration Enoxaparin, n/N (%)	Placebo, n/N (%)	Absolute Risk Difference (CI), %*
VTE at day 28†			
Women	23/1237 (1.9)	57/1247 (4.6)	-2.71 (-4.15 to -1.28)
Level 1	7/551 (1.3)	28/512 (5.5)	-4.20 (-6.46 to -1.94)
Level 2	16/680 (2.4)	28/728 (3.8)	-1.49 (-3.36 to 0.38)
Men	38/1248 (3.0)	43/1263 (3.4)	-0.36 (-1.79 to 1.07)
Level 1	18/519 (3.5)	19/528 (3.6)	-0.13 (-2.45 to 2.19)
Level 2	20/725 (2.8)	24/731 (3.3)	-0.52 (-2.35 to 1.30)
Age >75 y	18/725 (2.5)	50/743 (6.7)	-4.25 (-6.45 to -2.04)
Level 1	5/299 (1.7)	24/314 (7.6)	-5.97 (-9.37 to -2.57)
Level 2	13/424 (3.1)	26/428 (6.1)	-3.01 (-5.91 to -0.11)
Age ≤75 y	43/1760 (2.4)	50/1767 (2.8)	-0.39 (-1.48 to 0.71)
Level 1	20/771 (2.6)	23/726 (3.2)	-0.57 (-2.34 to -1.19)
Level 2	23/981 (2.3)	26/1031 (2.5)	-0.18 (-1.57 to 1.22)
Major bleeding events‡			
Women	14/1508 (0.9)	4/1511 (0.3)	0.66 (0.11 to 1.21)
Level 1	5/673 (0.7)	2/656 (0.3)	0.44 (-0.34 to 1.21)
Level 2	9/829 (1.1)	2/848 (0.2)	0.85 (0.07 to 1.63)
Men	11/1467 (0.75)	6/1477 (0.41)	0.34 (-0.20 to 0.89)
Level 1	4/619 (0.6)	0/625 (0.0)	0.65 (0.01 to 1.28)
Level 2	7/843 (0.8)	6/848 (0.7)	0.12 (-0.71 to 0.96)
Age >75 y	6/878 (0.7)	4/903 (0.4)	0.24 (-0.46 to 0.94)
Level 1	4/376 (1.1)	2/401 (0.5)	0.57 (0.68 to 1.81)
Level 2	2/500 (0.4)	2/501 (0.4)	0.00 (-0.78 to 0.78)
Age ≤75 y	19/2097 (0.9)	6/2085 (0.3)	0.62 (0.15 to 1.08)
Level 1	5/916 (0.5)	0/880 (0.0)	0.55 (0.07 to 1.02)
Level 2	14/1172 (1.2)	6/1195 (0.5)	0.69 (-0.05 to 1.43)

VTE = venous thromboembolism. \* For VTE end points in the total population, we report 95.8% CIs (P < 0.042) because of the  $\alpha$  adjustment for the interim analysis. For all other end points, we report 95% CIs (P < 0.050). † Assessed in the efficacy population (4995 patients). We excluded 21 patients

from the immobility subgroups because of missing immobility-level data or ab-sence of immobility-level classification. Of these, 1 had VTE (a woman >75 years who received placebo).

Assessed in the safety population (5963 patients). We excluded 22 patients because of missing immobility-level data or absence of immobility-level classification; we observed no major bleeding events among these patients. A threshold decrease in hemoglobin level of 20 g/L was incorporated as a post hoc criterion. Appendix Table 7 reports incidence of major bleeding events with the original threshold criterion of a 30-g/L decrease in hemoglobin level.

# Appendix Table 7. Incidence of Major Bleeding in Patient Subgroups, Using the Protocol-Defined Hemoglobin Decrease Threshold of 30 g/L\*

Subgroup	Incid	ence of Major Bl	eeding Events
	Extended- Duration Enoxaparin, n/N (%)	Placebo, n/N (%)	Absolute Risk Difference (CI), %†
Age or sex			
Women	13/1508 (0.9)	2/1511 (0.1)	0.73 (0.23 to 1.23)
Level 1	4/673 (0.6)	0/656 (0.0)	0.59 (0.01 to 1.18)
Level 2	9/829 (1.1)	2/848 (0.2)	0.85 (0.07 to 1.63)
Men	7/1467 (0.5)	5/1477 (0.3)	0.14 (-0.32 to 0.60)
Level 1	2/619 (0.3)	0/625 (0.0)	0.32 (-0.12 to 0.77)
Level 2	5/843 (0.6)	5/848 (0.6)	0.00 (-0.73 to 0.73)
Age >75 y	5/878 (0.6)	2/903 (0.2)	0.35 (-0.24 to 0.93)
Level 1	3/376 (0.8)	0/401 (0.0)	0.80 (-0.10 to 1.70)
Level 2	2/500 (0.4)	2/501 (0.4)	0.00 (-0.78 to 0.78)
Age ≤75 y	15/2097 (0.7)	5/2085 (0.2)	0.48 (0.06 to 0.89)
Level 1	3/916 (0.3)	0/880 (0.0)	0.32 (-0.04 to 0.70)
Level 2	12/1172 (1.0)	5/1195 (0.4)	0.61 (-0.08 to 1.29)
Age and sex Age >75 y			
Women	4/493 (0.8)	1/549 (0.2)	0.63 (-0.24 to 1.50)
Level 1	2/227 (0.9)	0/250 (0.0)	0.88 (-0.33 to 2.10)
Level 2	2/265 (0.8)	1/298 (0.3)	0.42 (-0.81 to 1.65)
Men	1/385 (0.3)	1/354 (0.3)	-0.02 (-0.77 to 0.73)
Level 1	1/149 (0.7)	0/151 (0.0)	0.67 (-0.64 to 1.98)
Level 2	0/235 (0.0)	1/203 (0.5)	-0.49 (-1.46 to 0.47)
Age ≤75 y			
Women	9/1015 (0.9)	1/962 (0.1)	0.78 (-0.17 to 1.39)
Level 1	2/446 (0.4)	0/406 (0.0)	0.45 (-0.17 to 1.07)
Level 2	7/564 (1.2)	1/550 (0.2)	1.06 (0.08 to 2.04)
Men	6/1082 (0.6)	4/1123 (0.4)	0.20 (-0.36 to 0.76)
Level 1	1/470 (0.2)	0/474 (0.0)	0.21 (-0.20 to 0.63)
Level 2	5/608 (0.8)	4/645 (0.6)	0.20 (-0.74 to 1.14)

VTE = venous thromboembolism. \* Assessed in the safety population (5963 patients). We excluded 22 patients because of missing immobility-level data or absence of immobility-level classification; we observed no major bleeding events among these patients.

<sup>+</sup> For VTE end points in the total population, we report 95.8% CIs (P < 0.042) because of the  $\alpha$  adjustment for the interim analysis. For all other end points, we report 95% CIs (P < 0.050).

Copyright of Annals of Internal Medicine is the property of American College of Physicians and its content may not be copied or emailed to multiple sites or posted to a listserv without the copyright holder's express written permission. However, users may print, download, or email articles for individual use.